Author's response to reviews

Title: Willingness to accept H1N1 pandemic influenza vaccine: A cross-sectional study of Hong Kong community nurses

Authors:

Samuel YS Wong (yeungshanwong@cuhk.edu.hk)
Eliza LY Wong (lywong@cuhk.edu.hk)
Josette Chor (josette@cuhk.edu.hk)
Kenny Kung (kungkl@ha.org.hk)
Paul KS Chan (paulkschan@cuhk.edu.hk)
Sian M Griffiths (siangriffiths@cuhk.edu.hk)
Carmen Wong (carmenwong@cuhk.edu.hk)

Version: 4 Date: 25 October 2010

Author's response to reviews: see over
25 Oct 2010

Dear editor,

On behalf of the first author and other co-authors, I am resubmitting the revised research article “Willingness to accept H1N1 Pandemic Influenza Vaccine: A Cross-sectional Study of Hong Kong Community Nurses” to BMC Infection Diseases.

We have revised the manuscript according to the editor’s suggestions by

- deleting the Addition file 1 from the system
- including the tables within the text
- removing all coloured text / underlining
- removing track changes

Thank you very much for publishing our paper in your journal which definitely gives our team a very strong encouragement.

Yours sincerely,

Eliza Wong (Corresponding author)
Assistant Professor
School of Public Health & Primary Care
The Chinese University of Hong Kong